Title: Heparin (GAG-hed) inhibits LCR activity of Human Papillomavirus type 18 by decreasing AP1 binding

Authors:

- Rita Villanueva (rvr1032000@yahoo.com.mx)
- Nestor Morales (nestor@correo.biomedicas.unam.mx)
- Irma Castelan (irmacast@ servidor.unam.mx)
- Enrique Garcia (enrique.villa@excite.com)
- Rocio Tapia (rociootp@fisio.cinvestav.mx)
- Angel Cid (cid@dkfz.de)
- Alejandro Garcia-Carranca (carranca@correo.biomedicas.unam)
- Esther Lopez-Bayghen (ebayghen@cinvestav.mx)
- Patricio Gariglio (vidal@cinvestav.mx)

Version: 5 Date: 28 August 2006

Author's response to reviews:

August 28th, 2006

Iratxe Puebla
Senior Assistant Editor
BMC-series journals
MS: 1883308808103186

Dear Prof. Puebla,

Please find enclosed the revised version of the manuscript entitled "Heparin (GAG-hed) inhibits LCR activity of Human Papillomavirus type 18 by decreasing AP1 binding".

Our answer to all request are as follow:

1. You requested to address the latest comments provided by the reviewer.
   We re-check tenses and we even ask to the American Journal Experts to help us with this matter. In figure legend 4 tense was corrected. The paragraph is now in past tense.
2. You have requested to go though the manuscript formatting checklist one more time and ensure that the revised manuscript conforms to all of the points. We did it carefully.
3. As you asked, we made the following changes:
   a. The image files are not including the title any more. It was done for all figures.
   b. Now, the figures are cropped as closely as possible.

We really appreciate your help and hope for this version to be complete and satisfactory to your entire request.

Sincerely,

Professor Esther Lopez-Bayghen, PhD